Author’s response to reviews

Title: A systematic review protocol on small/kiddie cigarette packaging size and its impact on smoking

Authors:

Halizah Mat Rifin (drhalizahmr@gmail.com)
Wan Shakira Rodzlan Hasani (wshakira@moh.gov.my)
Miaw Yn Ling (jane@moh.gov.my)
Tania Gayle Robert Lourdes (drtania@moh.gov.my)
Nur Liana Abd Majid (dr.liana@moh.gov.my)
Thamil Arasu Saminathan (thamilarasu.s@moh.gov.my)
Ahzairin Ahmad (ahz76@moh.gov.my)
Hasimah Ismail (hasimah.i@moh.gov.my)
Muhammad Fadhli Mohd Yusoff (fadhli_my@moh.gov.my)

Version: 4 Date: 28 Sep 2019

Author’s response to reviews:

Halizah binti Mat Rifin
Institute for Public Health,
National Institutes of Health,
Ministry of Health Malaysia
No. 1, Jalan Setia Murni U13/52,
Seksyen U13, Setia Alam,
40170 Shah Alam,
Selangor
halizah.matrifin@moh.gov.my
27th September 2019

Dear Editor-in-chief,

Thank you very much for your comments. We are thankful to the editors and reviewers for their time and insightful comments on our manuscript. We have amended their comments and suggestions and hope to submit a revised version of the manuscript for further consideration in the journal. Changes in the initial version of the manuscript are either highlighted for added sentences or strikethrough for deleted sentences in the revised version. Below, we also include a point-by-point feedback explaining how we have addressed each of the editors or reviewers’ comments as stated in Response Letter to Editor (SR Protocol Feedback on BMC SR 20_9_19). We look forward to the outcome of your assessment.

On behalf of the co-authors

Sincerely,

Dr Halizah binti Mat Rifin
Medical Officer
Institute for Public Health,
National Institute of Health,
Ministry of Health Malaysia